Author’s response to reviews

Title: Hsa-miR-125a-3p and hsa-miR-125a-5p are downregulated in non-small cell lung cancer and have inverse effects on invasion and migration of lung cancer cells.

Authors:

Lily Jiang (lilijiang1980@hotmail.com)
Qin Huang (qinh002018@tom.com)
Siyang Zhang (zhangsiyang_cmu11032@126.com)
Qingfu Zhang (qingfu1981@126.com)
Jihong Chang (jihongchang1981@163.com)
Xueshan Qiu (qiuxues@hotmail.com)
Enhua Wang (wangeh@hotmail.com)

Version: 2 Date: 30 March 2010

Author’s response to reviews: see over